Author's response to reviews

Title: Treatment of Renal Angiomyolipoma: Pooled Analysis of Individual Patient Data

Authors:

Teele Kuusk (teele.kuusk@gmail.com)
Fausto Biancari (fausto.biancari@ppshp.fi)
Brian Lane (blaneppmd@yahoo.com)
Conrad Tobert (conrad.tobert@gmail.com)
Steven Campbell (CAMPBES3@ccf.org)
Uri Rimon (Uri.Rimon@sheba.health.gov.il)
Vito D’Andrea (vito.dandrea@uniroma1.it)
Aare Mehik (aare.mehik@oulu.fi)
Markku H Vaarala (markku.vaarala@oulu.fi)

Version: 2
Date: 21 May 2015

Author's response to reviews:

Dear Editor,

Attached please find a revised version of our manuscript.

Following changes have been made, as required:
1) In the title page, the email address of all the authors of the manuscript is provided.
2) Line numbering and page numbers have been added.
3) Ethics statement has been added (lines 62-65). According to the Finnish law, no ethics council approval is required for retrospective chart reviews.
4) A statement concerning informed consents has been added (lines 65-66).

Sincerely, Markku Vaarala